Author's response to reviews

Title: A retrospective analysis of breast cancer subtype based on ER/PR and HER2 status in Ghanaian patients at the Korle Bu Teaching Hospital, Ghana

Authors:

Bernard Seshie (seshiegh2000@yahoo.com)
Nii A Adu-Aryee (naduaryee@chs.edu.gh)
Florence Dedey (flodedey@yahoo.com)
Benedict N.L Calys-Tagoe (calys75@hotmail.com)
Joe N Clegg-Lamptey (clegglamptey@chs.edu.gh)

Version: 5
Date: 25 June 2015

Author's response to reviews: see over
25th June 2015

The Editor
BioMed Central

Dear Editor,

MS: 1245936889142027 - A retrospective analysis of breast cancer subtype based on ER/PR and HER2 status in Ghanaian patients at the Korle Bu Teaching Hospital, Ghana.

I wish to submit the revised manuscript of the above paper.

We have addressed the requested concerns as follows:

1. I would like to see a remark about BRCA1/2 mutation status in the studied population and possible bias caused in the result

   This has been done.

2. TN patients with BRCA1 mutations or BRCA like tumors can profit from PARP inhibition treatment, please mention in the discussion.

   It has now been mentioned in the Discussion

3. We have made further language corrections

We have also made the following changes:

4. Objectives has been changed to Background in the Abstract
5. All Tables have been moved to the end of the manuscript after the References, and before the Figure Legends.
6. Duplicated figures have been removed
7. We have removed ‘et al’ from the References to conform to the Journal’s style

J. Clegg-Lamptey
For authors